Author’s response to reviews

Title: Developing Outcome, Process and Balancing Measures for an Emergency Department Longitudinal Patient Monitoring System using a Modified Delphi

Authors:

Marie Elizabeth Ward (m.ward@ucd.ie)
Abel Wakai (AWakai@rcsi.ie)
Ronald McDowell (ronaldmcdowell@rcsi.ie)
Fiona Boland (fionaboland@rcsi.ie)
Eoin Coughlan (eoincoughlan@yahoo.com)
Moayed Hamza (Moayed.hamza@ucdconnect.ie)
John Browne (j.browne@ucc.ie)
Ronan O'Sullivan (rgosullivan@bonsecours.ie)
Una Geary (ugeary@stjames.ie)
Fiona McDaid (fiona.mcdaid@hse.ie)
Éidín Ni Shé (eidin.nishe@ucd.ie)
Frances Drummond (francesjdrummond22@gmail.com)
Conor Deasy (Conor.Deasy@hse.ie)
Eilish McAuliffe (Eilish.mcauliffe@ucd.ie)

Version: 3 Date: 26 Nov 2018

Author’s response to reviews:

- If consent to participate statements are already present within the main text/Methods section that gives the above information, please also copy/place it under the 'Ethics approval and consent to participate' sub-section.
The methods section on invitation to participate in Delphi has now been replicated under “Ethics approval and consent to participate”

-- In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript. A statement has been inserted clarifying that the funding bodies had no part in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

-- Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship).

Currently, the contributions of all authors do not automatically qualify them for authorship. Please provide clarification on their contributions, or remove their names from the list of authors and place them in the “Acknowledgements” section instead. Please note that any changes in the list of authors requires the completion of the “Change in authorship” form.

Acquisition of funding, collection of data, writing of the manuscript or general supervision of the research group - alone - does not usually justify authorship.

-- We also notice that authors FD & ENS are missing from the authors' contributions section. Please could you include the contributions of all authors.

The authorship contribution section has been revised to include more detail on each authors contribution to the study. The authorship remains the same as the previous submission.

-- Figure files should contain only the image/graphic, as well as any associated keys/annotations. If titles/legends are present within the figure files, please remove them. Titles and legends have been removed from the figures.